Author’s response to reviews

Title: FORUM THEATER STAGING OF DIFFICULT ENCOUNTERS WITH PATIENTS TO INCREASE EMPATHY IN STUDENTS: DETERMINANTS OF EFFICACY. EVALUATION AT ANGERS UNIVERSITY MEDICAL SCHOOL

Authors:
Marion Sevrain-Goideau (mariongoideau@gmail.com)
Benedicte Gohier (begohier@chu-angers.fr)
William Bellanger (william.bellanger@univ-angers.fr)
Cedric Annweiler (cedric.annweiler@univ-angers.fr)
Mario Campone (mario.campone@univ-angers.fr)
Regis Coutant (recoutant@chu-angers.fr)

Version: 2 Date: 04 Feb 2020

Author’s response to reviews:

Dear Editor, Dear Reviewers,

We thank you for your positive review and the advised changes for improving the manuscript.

We have made the changes in the revised version of the manuscript.

Editor Comments:
1. We have now changed the title as follows
FORUM THEATER STAGING OF DIFFICULT ENCOUNTERS WITH PATIENTS TO INCREASE EMPATHY IN STUDENTS: DETERMINANTS OF EFFICACY. EVALUATION AT ANGERS UNIVERSITY MEDICAL SCHOOL
2. All the comments have been put in the appropriate box, and the revised submission is in the final form for publication.

Cindy Gallois (Reviewer 1)
We thank you for your positive review and for your help in improving the manuscript.
Tony Young (Reviewer 2):

If I am correct, you wish to get all comments in the reviewers’ box in the editorial manager tool. I’m not sure I have understood, as this is what we have done when we submitted the revised manuscript R1. I’m sorry if I don’t understand your comment.
I wonder whether you maybe wish to have all the Tables in the main manuscript, rather than as supplementary files.
This is what we have done in the revised version (R2) of the manuscript,
I hope this is what you suggested to do. If not, we would be please to perform the appropriate changes in the manuscript.